Author's response to reviews

Title: Fluid management guided by a novel continuous non-invasive arterial pressure device is associated with decreased postoperative morbidity in patients undergoing total knee and hip replacement - a randomized study

Authors:

Jan Benes (benesj@fnplzen.cz)
Lenka Haidingerova (haidingeroval@fnplzen.cz)
Jiri Pouska (pouskaj@fnplzen.cz)
Jan Stepanik (stepanikj@fnplzen.cz)
Alena Stenglova (simanovaa@fnplzen.cz)
Jan Zatloukal (zatloukalj@fnplzen.cz)
Richard Pradl (pradl@fnplzen.cz)
Ivan Chytra (chytra@fnplzen.cz)
Eduard Kasal (kasal@fnplzen.cz)

Version: 2
Date: 16 December 2014

Author's response to reviews:

Previous letter to the editor is valid, only minor formating changes were provided in the current version of the manuscript.